Author's response to reviews

Title: Preliminary study of p53 and c-erbB-2 expression in gallbladder cancer in Indian patients

Authors:

Amita Chaube (amitachaube@yahoo.com.in)
Mallika Tewari (mallika_vns@satyam.net.in)
R S Garbyal (rajgarbyal@yahoo.co.in)
Usha Singh (harishukla@usa.net)
Hari S Shukla (harishukla@usa.net)

Version: 9 Date: 26 April 2006

Author's response to reviews: see over
Title:
Preliminary study of p53 and c-erbB-2 expression in gallbladder cancer in Indian patients

Authors:

Amita Chaube (amitachaube@yahoo.com.in)
Mallika Tewari (mallika_vns@satyam.net.in)
Rajendra S Garbyal (rajgarbyal@yahoo.co.in)
Usha Singh (harishukla@usa.net)
Hari S Shukla (harishukla@usa.net)

Date: 26 April 2006

Dear Sir,

I am grateful to you for accepting our paper for publication in your journal.

I have made the necessary additions as suggested in the text (viz. approval of ethics committee and informed consent from patients).

Please let me know if any further corrections are needed.

Thanking you,
Yours sincerely,

Hari S Shukla
IMS, BHU
Varanasi (INDIA)